

# Correlates of eating disorder pathology in Saudi Arabia: BMI and body dissatisfaction.

Melisse, B.; Blankers, M.; Beurs, E. de; Furth, E.F. van

# Citation

Melisse, B., Blankers, M., Beurs, E. de, & Furth, E. F. van. (2022). Correlates of eating disorder pathology in Saudi Arabia: BMI and body dissatisfaction. *Journal Of Eating Disorders*, *10*(1), 126. doi:10.1186/s40337-022-00652-4

Version:Publisher's VersionLicense:Creative Commons CC BY 4.0 licenseDownloaded from:https://hdl.handle.net/1887/3515243

Note: To cite this publication please use the final published version (if applicable).

# RESEARCH

**Open Access** 

# Correlates of eating disorder pathology in Saudi Arabia: BMI and body dissatisfaction



Bernou Melisse<sup>1,3,4\*</sup>, Matthijs Blankers<sup>2</sup>, Edwin de Beurs<sup>2,3</sup> and Eric F. van Furth<sup>4,5</sup>

## Abstract

**Background:** Saudi Arabia is undergoing rapid sociocultural changes, which may have led to an increase of body mass index and eating disorder pathology. The aim of this study is to investigate whether body dissatisfaction, self-esteem, having lived abroad, cultural orientation, perceived stress, media use, and socioeconomic status are correlates of eating disorder pathology with body mass index as a covariate. Additional aims are to investigate if cultural orientation is associated with symptomatology and if stress is a covariate in the association between eating disorder pathology and Western orientation.

**Method:** Self-report measures were administered in a convenience Saudi community sample (N = 1225) between April 2017 and May 2018. Hierarchical multiple linear regression analyses with eating disorder pathology as dependent variable were performed to establish the associations among the variables.

**Results:** After adjusting for the effect of BMI, only body dissatisfaction and eating disorder pathology were moderately associated. Eating disorder pathology and body dissatisfaction were more severe among Saudi citizens with a higher BMI.

**Discussion:** Several explanations for the lack of associations of westernization, self-esteem, and stress with eating disorder pathology are reviewed and discussed. The majority of this convenience sample existed of young unmarried Saudi females of high socioeconomic status. Of the total sample, 35% displayed eating disorder pathology which may be a reflection of the high rates of excess weight.

Keywords: Eating disorder pathology, BMI, Body dissatisfaction, Saudi Arabia

### **Plain English summary**

Eating disorders were assumed to be rare in Saudi Arabia, however, Saudi citizens report both, increased pressure to be thin and eating disorder pathology. Saudi Arabia is dealing with rapid sociocultural changes. Consequences of these sociocultural changes, which might be associated with eating disorder pathology, are an increase of BMI, body dissatisfaction, low self-esteem, westernization, perceived stress, media use, and socioeconomic status. In addition, the increase in BMI might strengthen the associations between eating disorder pathology and body dissatisfaction, low self-esteem, perceived stress, media use, and socioeconomic status. Alternative explanations might be that cultural values are associated with different eating disorder symptoms or that eating disorder pathology is associated with stress stemming from transformation that aims to empower women and modernize Saudi society. This study

\*Correspondence: bernoumelisse@outlook.com

<sup>1</sup> Novarum Center for Eating Disorders & Obesity, Laan van de Helende Meesters 2, 1186 AM Amstelveen, The Netherlands

Full list of author information is available at the end of the article



© The Author(s) 2022. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

showed that only BMI and body dissatisfaction were associated with eating disorder pathology, and BMI was a covariate in the association between eating disorder pathology and body dissatisfaction. Preventative programs focused on avoidance of maladaptive weight loss strategies and the improvement of body satisfaction might be beneficial in Saudi Arabia.

#### Background

Eating disorders have historically been associated with Caucasian women in developed Western countries [1]. Therefore, eating disorders have been perceived as culture bound syndromes [2], and most research on eating disorders is conducted in Western countries [3]. However, cultural factors are essential to understand the etiology of eating disorders [4]. In Western countries, the main feature of eating disorders is the desire to be thin [1]. In Saudi Arabia traditional notions of beauty are different, with a curvy body ideal being associated with fertility and wealth. Thus, Saudi citizens are thought to be less likely to suffer from eating disorders [5]. However, though data are still inconclusive, the thin ideal seems on the rise in Saudi Arabia [6-8]. Recent studies have shown that eating disorders occur globally [9, 10] and eating disorders occur particularly in cultures in transition as they tend to adopt Western values [2, 11], illuminating the interplay between culture and psychopathology [12]. This is especially relevant to Saudi Arabia, as it is undergoing rapid sociocultural changes [6, 13].

Rapid sociocultural changes are often referred to as a combination of westernization, industrialization and globalization [1, 14, 15]. Westernization is defined as adoption of language, lifestyle, values and beliefs of Western cultures [14]. Western oriented Arabs display greater assimilation with Western countries than those who are more Arab oriented, resulting in Saudi citizens speaking English with their relatives and friends, and finding entertainment in malls rather than in a traditional souk [16]. In addition, industrialization comes along with an increased sedentary lifestyle, and a dietary shift towards Western types of foods, higher in salt, sugar and fat, all instrumental in the rise of excess weight [17, 18]. The prevalence of high BMI has increased from below 10% around 1980 to rates of 29-58%, and Saudi Arabia currently has one of the highest prevalence rates of excess weight [18-20]. Excess weight is associated with eating disorder pathology [21], as Saudi citizens with excess weight are known to be 2-3 times more likely to develop an eating disorder pathology than Saudi citizens without excess weight [22]. Furthermore, excess weight is associated with body dissatisfaction and low self-esteem [23]. Moreover, studies found that high BMI strengthens the association of eating disorder pathology with: body dissatisfaction, low self-esteem, perceived stress, and media use [17, 22, 24]. High BMI may serve as a covariate in the interplay between eating disorder pathology and its correlates.

Other consequences of westernization include an increase in media use, which may be associated with eating disorder pathology [3, 25]. Media use is defined as the use of internet, social media platforms, and streaming services [26]. Saudi citizens are extensive media users, internet access has increased over 100-fold in the last 16 years, and, on average, they spend 2.7 h daily watching streaming services, 50-100% more than in Western countries [27-29]. In addition, social media photos and videos often promote a more Western presentation of shape and weight, potentially resulting in increased thin ideal and consequently unhealthy dietary behavior [30]. Furthermore, various diets are promoted on social media [31]. However, the exact impact of westernization on the development of eating disorder pathology is still inconclusive. Some studies show that a Western orientation and having lived abroad, appeared to be risk factors for eating disorder pathology in several Gulf countries, and Gulf citizens who live in a Western country display more severe eating disorder pathology than their counterparts who still live in their country of origin [11, 22, 32–35]. In contrast, some studies show that internalization of Western values is weakly associated with eating disorder pathology in Saudi Arabia [36]. Finally, cultural orientation might be associated with symptomatology: individuals who are predominantly Arab orientated engage more in restrictive eating behaviors, while individuals who are more westernized display more binge eating behaviors [35, 37]. The role of religion may go both ways, as the Holy Quran supports restrictive eating stating that someone should fill its body 1/3 with air, 1/3 with water and 1/3 with food. Furthermore, restrictive eating during the holy month of Ramadan does not affect eating behavior [38–40]. On the other hand, social events are culturally accompanied by the intake of large amounts of food [41].

The contradictory results regarding westernization suggest that, rather than just westernization, conflicting cultural values related to the sociocultural, political and legal transformations that aim to empower women and modernize the relatively conservative Saudi society might cause stress [22, 42]. Maladaptive coping mechanisms to respond to stress may contribute to the development of eating disorders, as stress is associated with binge eating, restraint eating and overeating [43]. The role of perceived stress should therefore be taken into consideration as well, when examining correlates of eating disorder pathology [44]. Lastly, the sociocultural changes led to an increase in socioeconomic status (SES) of Saudi inhabitants since a free educational system is developed [6]. Consequently, level of education and employment rates of Saudi citizens increased [32, 45]. However, the exact impact of SES on eating disorder pathology is unclear: in Saudi Arabia SES appears to be associated with restrictive eating behavior [32], which is confirmed by studies conducted in Western countries. However, SES appears to be negatively associated with excess weight, whereas binge eating disorder appears unrelated to SES in Western societies [1, 46, 47]. Lastly, the increase in SES is associated with larger families and therefore an increase in youth. Median age in Saudi Arabia is 31.8 years and about 72% is aged between 15 and 64 years [48, 49]. The growth of the adolescent population is potentially associated with an increase in eating disorder pathology as the prevalence appears to increase among adolescent Saudi citizens [50-53].

In sum, as Saudi Arabia is undergoing rapid sociocultural changes, several risk factors for the development of eating disorder pathology may currently apply. Though studies have identified correlates of eating disorder pathology in Western countries, few have been undertaken in Saudi Arabia [26, 54]. Once correlates of eating disorder pathology are identified targeted preventative programs can be initiated. The aim of this cross-sectional explorative study was to examine whether body mass index (BMI), body dissatisfaction, self-esteem, cultural orientation, perceived stress, media use, and SES are correlates of eating disorder pathology in Saudi Arabia. Additional aim is to investigate if a high BMI is a covariate in these potential associations. It is hypothesized that, eating disorder pathology is positively associated with greater body dissatisfaction, low self-esteem, Western cultural orientation rather than Arab orientation, increased levels of perceived stress, more time spend on media use, and increased SES, and that a larger BMI serves as a covariate in these associations. An alternative hypothesis is that a greater level of stress is a covariate in the association between eating disorder pathology and Western orientation. An additional alternative hypothesis is that cultural orientation is associated with symptomatology, where Saudi citizens who are Arab oriented show more restrictive behavior and those with who are Western oriented more binge eating behavior.

# Methods

#### Procedure

A convenience sample was recruited between April 2017 and May 2018 from students (Princess Noura bint Abdulrahman University (PNU), King Saud University, Sixth High School for Quran Memorization) in Riyadh, and through social media (Twitter, Facebook, Snapchat). Several sports facilities (NuYu gym, Sukoun yoga studio) in Riyadh, Dammam and Jeddah also shared a link among their members. Participants were also recruited through the first authors (BM) social network. Some of BMs students recruited participants through their personal network. Inclusion criteria were being a Saudi passport holder, literate and age  $\geq$  18 years old. The aim was to reach as many Saudi citizens as possible. As Saudi society is being perceived as a relatively culturally reclusive society [55], it is hard to access its citizens, and because eating disorder pathology has only been poorly studied, data collection involved targeting Saudi citizens in the country, which has resulted in a unique sample. All participants (N=1229) were Saudi nationals, however four participants (3.2%) had  $\geq$  5% missing data with regard to measure items and were therefore excluded. Participants (N=1225) were literate, and aged between 18–81 years old, mean age 23.6 (SD = 8.5) years (Table 1). Participants provided informed consent and completed anonymously an online self-report survey through Survey Monkey. Questions about the study could be sent by email to BM.

#### Measures

#### Eating Disorder Examination-Questionnaire

The Saudi version of the Eating Disorder Examination-Questionnaire (EDE-Q) was used to assess eating disorder pathology during the last 28 days [19, 56]. The EDE-Q is a self-report instrument of 28 items, resulting in a global score indicating the severity of eating disorder pathology and additionally measures frequency of eating disorder behaviors such as binge eating episodes. Severity of eating disorder pathology was measured on a 7 point- Likert scale (0: feature was absent, to 6: feature was markedly present or present every day) [56]. The EDE-Q was translated to Arabic and some cultural adjustments were made in the Saudi EDE-Q e.g., communal changing rooms and pools are rare and were replaced by gyms and weddings, as further explained in Melisse et al. (2021) [19]. The Saudi version has good psychometric properties [19]. However, the factor structure with four subscales has not been confirmed with factor analysis. Therefore, only the EDE-Q global score was used. The Saudi EDE-Q has a Cronbach's  $\alpha$  of 0.89, clinical cut-of is 2.93 [19]. Cronbach's  $\alpha$  in this study was 0.90.

#### Table 1 Demographics

	Ν	MD (S)
Age	1225	23.6 (8.5)
18–25 years	919 (75.0%)	
26–40 years	228 (18.6%)	
41–65 years	70 (5.7%)	
66–81 years	3 (0.2%)	
Gender		
Unknown	11 (0.9%)	
Female	1048 (85.6%)	
Male	166 (13.6%)	
BMI	1225	25.2 (7.0)
Marital status	1225	
Unknown	1 (0.1%)	
Married	211 (17.2%)	
Unmarried	938 (76.5%)	
Divorced	75 (6.1%)	
Occupation/education	1225	
Unknown	15 (1.2%)	
High school	315 (25.7%)	
University in country of heritage	474 (38.7%)	
University in Arab country	25 (2.0%)	
University in Western country	6 (0.5%)	
Employed	182 (14.9%)	
Unemployed	127 (10.4%)	
Other	81 (6.6%)	
Self-report measures		
EDE-Q	1225	2.7 (1.4)
BSQ	1225	86.5 (36.3
Rosenberg self-esteem	1220	27.0 (79.4
ARMSA Western orientation	1222	1.1 (0.6)
ARMSA Arab orientation	1115	1.4 (0.5)
Perceived stress scale	1003	27.5 (71.8
Eating disorder behaviors <sup>†</sup> , reported by <i>N</i> , (%)		
Objective binges	1078 (88.0%)	6.8 (3.4)
Self induced vomiting	392 (32.0%)	2.5 (6.0)
Laxative misuse	401 (32.7%)	3.0 (0.6)
Extensive exercise	967 (78.9%)	8.7 (9.8)

ARMSA Acculturation Scale for Mexican Americans, BMI Body Mass Index, BSQ Body Shape Questionnaire, EDE-Q Eating Disorder Examination-Questionnaire

<sup>+</sup> Frequency of eating disorder behaviors reported during the last 4 weeks

#### Body Shape Questionnaire

The Body Shape Questionnaire (BSQ), measures the severity of body dissatisfaction, such as fear of gaining weight, desire to lose weight and physical- appearance related self-devaluation during the last 28 days. The BSQ is a self-report measure of 34 items on a 6 point- Likert scale (1: never, to 6: always) [57]. The overall score is the total of all items and can range between 34 and 204, higher scores indicate greater body dissatisfaction.

Scores  $\geq$  80 indicate mild concern,  $\geq$  111 moderate concern and  $\geq$  140 marked concern with shape. The BSQ has a Cronbach's  $\alpha$  of 0.96, a high test–retest reliability and good concurrent validity [58]. BM and a clinical psychology student of PNU slightly adapted an Arabic version of Mousa et al. (2010) with regard to the Arabic language [59]. In addition, one cultural adaptation was made in question 27: as females in Saudi Arabia share cars rather than travel by bus, "bus seat" was changed to "car seat". Cronbach's  $\alpha$  in this study was 0.96.

#### Rosenberg self-esteem scale

The Rosenberg self-esteem scale measures self-esteem with 10 items representing statements with a 4-point Likert response scale (1: strongly disagree, to 4: strongly agree), five of these items (2, 5, 6, 8, 9) were reverse scored. Higher scores indicate higher self-esteem and sum scores can range from 10 to 40 [60]. Cronbach's  $\alpha$  of the Arabic version was 0.71, the instrument has a high test–retest reliability and good concurrent validity [61–63]. Cronbach's  $\alpha$  was 0.70 in the present study.

#### Acculturation Rating Scale for Mexican-Americans II

The revised version of the Acculturation Rating Scale for Mexican-Americans (ARMSA II) [64, 65] is a westernization survey that measures two unrelated dimensions: maintenance of the original culture and identification with the recently introduced culture, originally researched among Mexican immigrants. Western and Mexican orientation were each assessed in 12 behaviorally based questions on a four point Likert scale (1: never to 4: very often) for language, media, food and consumer goods. Total scores on both cultural orientation scales ranged between 0 and 36, scores on both dimensions can be used separately or deducted [64]. Adaptations were made similar to those of Stigler et al. (2010) who adapted the ARMSA II for use among Indians in their country of origin. In the adapted version Cronbach's alphas were 0.84 for the Indian scale and 0.90 for the Western scale [66].

The ARMSA II was translated to Arabic by a native speaker, a clinical psychology student of PNU, with a parallel translation by a professional translator. Minor differences between both versions were discussed and resolved, and a backtranslation was performed. The ARMSA II was adjusted, resulting in an Arab orientation and Western orientation scale. The following cultural adaptations were made by BM: language preferences involving Arabic as mother tongue, English as Western language, media preferences as Arabic versus Western shows/films/music. With regard to food/restaurants, common Western coffee houses in Saudi Arabia such as Java Time, Starbucks and Dr. Cave, and common Arabic restaurants such as Najd Village and Isteraha (desert-camp), and traditional Saudi dishes such as kabsah, maqluba, dates, samboosa and dolmah, were added. Traditional markets were mentioned as souks, as were traditional clothes, jallabiyah (for women) and thoob (for men) respectively. Cronbach's alphas were 0.82 for the Arab scale and 0.90 for the Western scale. Scores on these scales were weakly correlated (Pearson's r=0.27, p < 0.01).

#### Perceived Stress Scale

The Perceived Stress Scale (PSS) measures subjective feelings of stress on 10 items using a 5-point Likert scale (0: never to 4: very often) during the last month. Positively stated items (4, 5, 7, 8) were reverse scored, total score range is 0–40. Higher scores indicate more severe feelings of stress [67]. The Arabic version has Cronbach's  $\alpha$  of 0.80 and high test–retest reliability [68]. Cronbach's  $\alpha$  in the present study was 0.83.

#### Media use

Participants completed questions about usage (time and frequencies) of internet, social media and streaming services as described in Fardouly and Vartanian (2014) [69]. They also provided information about use of streaming services, provided in Additional file 1: Supplementary Table A. Internet and streaming services use were measured with a single item. Cronbach's  $\alpha$  was 0.80 for social media and 0.78 for overall media use in the present sample.

#### **Body Mass Index**

BMI was calculated as weight in kg/squared height in meters as reported in the EDE-Q. WHO guidelines were applied to classify BMI [70].

#### Socioeconomic status and demographic data

To estimate SES, questions about level of education, parental level of education, level of profession (e.g. manual, managerial) and parental level of profession were administered. Answers to these questions were administered as a Likert scale (e.g. level of profession 1: housewife/ unemployed, to 4: professional). Housewife was indicated as low SES since housewives in Saudi Arabia are generally lower educated and women of high SES are usually employed and can afford nannies, maids and drivers [32, 45, 71]. Each item received a numeric score of which all items together resulted in a total score, global scores were averaged item scores. Higher global scores represented higher SES. Demographic data such as gender, age, marital status, nationality, and history of living abroad were also requested from the participants.

#### Statistical analysis

To test for multicollinearity between independent variables correlation analyses were performed. Differences in eating disorder pathology were examined by ANOVA for continuous variables (age) and chi-square for non-continuous variables (gender, marital status and education/ profession). All data met the assumptions of normality, homoscedasticity, and linearity. A hierarchical multiple linear regression analysis, forward stepwise method was performed with body dissatisfaction, low self-esteem, having lived abroad for at least six months, cultural orientation, perceived stress, media use, and socioeconomic status as independent variables and eating disorder pathology as dependent variable. In step one, BMI was entered to control for its potential cofounding effect; in step 2, the independent indices were allowed to enter, to test whether they would predict additional variance in eating disorder pathology, as reflected in the EDE-Q total score, over BMI. In addition, a Bonferroni post-hoc analysis was performed. A one-way ANOVA was used to test if severity of eating disorder pathology and body dissatisfaction significantly differed among the various BMI groups. Data were analyzed in SPSS version 25.

#### **Ethical considerations**

Ethical approval was given on May 7th, 2017 (17-0097), by the ethical review board of PNU in Riyadh, Saudi Arabia.

#### Results

All participants (N=1225) were Saudi nationals, literate, and aged between 18–81 years old, mean age 23.6 (SD=8.5) years (Table 1). There were several differences between the study sample compared to the general Saudi population: females were overrepresented, 85.6% (n=1048) versus 42.3%, the majority was single (76.5% vs. 33.0%) and a smaller percentage married (17.2% vs. 58.8%) [72]. The study sample was also more highly educated: most participants attended university (41.2% vs. 4.4%) [73], and 14.9% was employed, compared to 30.2% generally [74]. Participants were from all over Saudi Arabia (details available from BM), though most resided in the larger cities. Post-hoc calculated power, considering 13 predictors and an observed  $R^2$  of 0.60,  $\alpha$ =0.05 (2 sided) was 100%.

#### Eating disorder pathology

There was no effect of age (*F*(62) = 4677.95; *p* = 0.510), gender ( $\chi^2(1) = 107.8$ ; *p* = 0.83) or employment/education

 $(\chi^2(6) = 718.18; p = 0.052)$  on the EDE-Q global score. There was an effect of marital status ( $\chi^2(2, 1220) = 209.55$ ; p = 0.002). Married Saudi citizens had less eating disorder pathology than unmarried Saudi citizens. Of the participants, 34.8% (n=426: 24.7%, n=41 male, 36.3%, n = 380 female) had an EDE-Q score above the clinical cut-off of 2.93 [19]. Multiple comparisons (F = 160.39, p < 0.001) showed that, based on BMI, the average scores on the EDE-Q were different between Saudi citizens with a BMI < 18.5, a BMI between 18.5 and 24.9 and BMI between 25.0 and 29.9 (p < 0.001). EDE-Q scores did not differ between Saudi citizens with a BMI between 25.0 and 29.9 or 30.0 and 39.9 (p = 0.369) and between Saudi citizens with a BMI between 30.0 and 39.9 or BMI>40 (p = 0.469). BSQ scores of Saudi citizens with BMI < 18.5 were comparable to those with BMI between 18.5 and 24.9 (p = 0.405), but differed for BMI between 25.0 and 29.9 (p < 0.001) (F = 149.52, p < 0.001). There were no differences in BSQ scores for BMI between 30.0 and 39.9 and BMI > 40 (p = 0.133) (Table 2). This suggests greater body dissatisfaction and eating disorder pathology among participants with a higher BMI.

#### Correlates of eating disorder pathology

Table 3 shows that there was no problematic multicollinearity between independent variables. Table 4 shows that in Step 1 of the hierarchical regression analysis, the association between BMI and eating disorder pathology was positive, statistically significant (p < 0.001) and substantial, accounting for 35.2% of the variance in eating disorder pathology. In Step 2, body dissatisfaction, selfesteem, having lived abroad, cultural orientation, perceived stress, media use, and socioeconomic status were allowed to enter. BMI and body dissatisfaction together explained 60% of the variance in eating disorder pathology. After adjusting for BMI, body dissatisfaction was moderately associated with eating disorder pathology, and explained 24.8% of the variance. BMI appeared to serve as a covariate. However, after entering body dissatisfaction no other variables contributed to explaining variance in eating disorder pathology. Together, all variables explained 64% of the variance ( $R^2 = 0.64$ , F = 41.14, p < 0.001). Bonferroni post-hoc analysis did not reveal different results as all significance levels were p < 0.003.

Pearson zero-order correlations were statistically significant but weak between EDE-Q total score and having lived abroad (r = 0.28, p < 0.001), use of internet (r = 0.28, p < 0.001), social media (r = 0.28, p < 0.001), and SES (r=0.38, p<0.001). The associations with self-esteem (p=0.598), Western orientation (p=0.991), Arab orientation (p=0.441), perceived stress (p=0.365), and streaming services (p = 0.598) were not statistically significant. In addition, when both scales of cultural orientation were adjusted for the effect of the other (e.g. Western orientation deducted from Arab orientation) no association was found with eating disorder pathology (p=0.090). The alternative hypothesis was not confirmed: perceived stress was not a covariate in the association between eating disorder pathology and westernization (p = 0.468). Linear regression analysis showed that there were no associations between Arab orientation and restrictive eating behavior as measured by the restraint scale of the EDE-Q (p = 0.297), neither between Western orientation and binge eating behavior as measured by the EDE-Q open ended question regarding number of binge eating episodes (p = 0.180).

#### Discussion

The main findings of this explorative cross-sectional study are that BMI, body dissatisfaction, lived abroad, media use, and SES were associated with eating disorder pathology, as was single marital status. However, when these correlates were assessed together only body dissatisfaction and BMI appeared to be associated with eating disorder pathology. The results of this study regarding body dissatisfaction and BMI are comparable to other

BMI	1. BM	< 18.5		2. BMI 18.5–2	between 4.9		3. BMI bet 25.0–29.9	ween	4. BM 30.0–	l between 39.9		5. BMI > 4	0
	n	%	<u> </u>	n	%		n	%	n	%		n	%
	131	1	1.0	562	47.3		281	23.7	281	14.6	6	40	3.4
		М	SD	М	SD	М	SD	М	SD	М	SD	Pairwis compa p<0.00	rison
EDE-Q total so	core	1.6	1.3	2.4	1.3	3.3	1.1	3.6	1.1	3.9	3.9	1<2<3	3 = 4 = 5
BSQ score		53.7	19.6	75.3	30.0	102.3	31.4	112.4	35.6	132.8	34.7	1=2<	3 <b>=</b> 4 <b>=</b> 5

Table 2 Means (and SDs) of EDE -Q and BSQ scores by five BMI groups and results of multiple comparisons of Tukey posthoc tests

BMI Body Mass Index, BSQ Body Shape Questionnaire, EDE-Q Eating Disorder Examination-Questionnaire

	BMI	Body dissatisfaction	Low self- esteem	Lived abroad	Western orientation	Arab orientation	Western– Arab orientation	Perceived stress	Streaming services	Internet	Internet Social media	Overall media use	SES
BMI	1.00												
Body dissatis- faction	0.27**	1.00											
Low self- esteem	0.02	0.04	1.00										
Lived abroad	0.02	0.02	0.23**	1.00									
Western orien- tation	- 0.07	0.06	0.06	0.75	1.00								
Arab orienta- tion	0.03	0.05	- 0.14	0.01	0.27**	1.00							
Western-Arab orientation	0.07	0.22	0.11	— 0.01	0.75	- 0.56	1.00						
Perceived stress	- 0.02	0.02	0.01	- 0.16	- 0.09	- 0.06	0.03	1.00					
Streaming services	0.45**	0.11**	- 0.04	0.36**	— 0.42**	0.13	- 0.20	- 0.03	1.00				
Internet	- 0.09	0.03	- 0.01	-0.14*	- 0.05	0.12	0.02	0.04	- 0.35**	1.00			
Social media	0.02	0.02	0.04	-0.23	0.06	0.04	0.001	0.04	- 0.36**	-0.35**	1.00		
Overall media use	0.02	0.02	- 0.04	- 0.23	— 0.06	- 0.06	0.08	0.04	0.36**	0.99**	1*	1.00	
SES	0.04	0.15*	0.35**	0.31	- 0.20*	- 0.08	-0.21	0.02	0.55**	0.18	0.38*	0.38*	1.00
<i>BMI</i> Body Mass Ir * <i>p</i> < 0.05 ** <i>p</i> < 0.001	ndex, SES so	BMI Body Mass Index, SES socio-economic status $*p < 0.05$											

 Table 3
 Pearson correlation matrix of independent variables

Table 4 Mult	iple hierarchical	regression wi	ith eating	disorder i	pathology	as dependent variable

	EDE-Q	total sco	re*				Correlation	s	95% confidence	ce interval
	β	t	p	F	df	$\Delta$ adjusted $R^2$	Zero order	Part r	Lower bound	Upper bound
Step 1										
BMI	0.12	6.41	< 0.001	139.65	1224	0.35	0.60	0.60	0.10	0.14
Step 2										
BMI	0.07	2.99	< 0.001	156.9	1115	0.35	0.60*	0.38	0.05	0.12
Body dissatisfaction	0.58	9.26	< 0.001	143.8	1115	0.60	0.75*	0.50	0.42	0.74
Self-esteem	0.003	0.53	0.598	60.3	1190	NA	0.28	0.04	- 0.01	0.12
Lived abroad for $\geq$ 6 months	0.169	0.10	0.221	66.8	1223	NA	0.28*	0.09	-0.104	0.44
Western orientation	0.003	0.01	0.991	0.1	1201	NA	- 0.02	0.001	- 0.50	0.51
Arab orientation	- 0.26	-0.83	0.410	1.3	1201	NA	-0.11	- 0.06	- 0.88	0.36
Western-Arab orientation	0.16	1.72	0.090	0.8	1201	NA	0.10	0.20	- 0.28	0.74
Perceived stress	0.02	1.04	0.302	0.8	1190	NA	0.03	0.08	- 0.02	0.06
Streaming services	- 0.02	-0.53	0.598	0.3	1223	NA	- 0.02	-0.01	- 0.07	0.04
Internet use	0.004	8.34	0.798	69.6	1223	NA	0.28*	- 0.02	- 0.05	0.07
Social media	- 0.02	0.61	0.796	68.5	1223	NA	0.28*	- 0.02	-0.14	0.11
Media use total	0.003	0.61	0.542	68.7	1223	NA	0.28*	0.05	- 0.01	0.13
Socioeconomic status	-0.11	-0.48	0.631	80.2	1223	NA	0.38*	0.04	- 0.05	0.03

BMI Body Mass Index, EDE-Q Eating Disorder Examination-Questionnaire

\*Significant at p < 0.001 level

studies [17, 22]. Body dissatisfaction was moderately associated with eating disorder pathology after adjusting for the effect of BMI. In line with other studies [75–77], body dissatisfaction was the strongest correlate of eating disorder pathology among various other correlates, and Saudi citizens with a high BMI had both, greater severity of eating disorder pathology and body dissatisfaction. Around 35% of this Saudi community sample had an EDE-Q score above clinical cut-off. Therefore, the high prevalence of eating disorder pathology may reflect the high rates of excess weight and maladaptive strategies to lose weight. This is of concern, as around half of the Saudi population has a BMI above 25 [19]. Preventative programs focused on avoidance of maladaptive weight loss strategies and the improvement of body satisfaction might be beneficial.

This study did not find an association between westernization and EDE-Q scores. Other studies found associations between westernization and EDE-Q scores. Failure of this study to reproduce the association between eating disorder pathology and westernization may be explained by the fact that only small associations were found in some other studies [11, 36]. In addition, the rapid sociocultural changes have also been associated with elevated levels of industrialization [11, 16]. Industrialization has coincided with an increase in mental health problems including eating disorders [11, 78]. In addition, societies dealing with faster industrialization displayed greater risk for eating disorders [16, 52]. Furthermore, other studies found that cultural orientation was associated with specific eating disorder pathology [22, 24, 35, 37, 45]. This alternative hypothesis was also not confirmed in current study: Arab orientation and restrictive eating behavior, and Western orientation and binge eating behavior were not associated. In addition, other studies suggested the role of stress in the association between eating disorder pathology and westernization [44]. Stress was not a covariate in the association between westernization and eating disorder pathology in current study. Finally, results of current study were not in line with other studies who found associations between eating disorder pathology and self-esteem [23], use of internet, social media and streaming services [3, 25, 26], and SES [32]. However, since the association of working status (housewife versus employed) and SES of mothers appeared inconclusive [79] the analysis was repeated eliminating housewife from the composite SES score. This analysis did not yield different results. In addition, other studies found BMI to serve as a covariate in the associations of eating disorder pathology and perceived stress, media use, and low self-esteem [17, 22, 24]. None of these findings were confirmed in the present study. Explanation might be that in these other studies small associations were found, most samples involved teenagers rather than adults, while severity of eating disorder pathology was inversely related to age [80, 81].

Several study limitations should be taken into consideration. First, BMI was calculated on self-reported weight

and height, and social desirability may have instigated participants to report lower weight than their actual weight [82]. Second, the composition of the convenience sample differed substantially from the general Saudi population regarding gender and education level. No effect on EDE-Q total score was found of age, gender and education/occupation. In addition, results of a convenience sample are less reliable than a clinical sample and potentially impacts generalizability [16, 19]. Third, the sample may also be biased by participants' readiness to report their eating disorder pathology. Factors not measured may also have caused selection bias: respondents may have participated because they are more interested in health care, mental health care, or eating disorder pathology, or because they have more concerns regarding their body shape or eating behavior compared to the general population [19]. Consequently, the results should be interpreted with care, as generalizability to the overall Saudi population is limited. At the same time, this study does provide information regarding the association of eating disorder pathology, BMI and body dissatisfaction. Preventative programs can be offered to young unmarried Saudi citizens with body dissatisfaction, as selective prevention programs display larger effects than universal prevention programs [83–86]. In addition, improved knowledge is likely to increase the number of people seeking treatment and decrease the stigma of psychotherapy [16].

Although the Acculturation Rating Scale for Mexican-Americans II (ARMSA II) appeared reliable, there were small limitations to its internal validity: the ARMSA II originally measured acculturation among migrants in the USA rather than acculturation among populations dealing with sociocultural changes in their country of origin. However, the ARMSA II was validated and used in international studies. In addition, the overall frequency and time Saudi citizens spent on (social) media was examined. Some studies found that greater emotional investment, negative feedback seeking and social comparison were associated with eating disorder pathology rather than time spent or number of actions on (social) media [26, 87]. Finally, as only associations among the variables could be investigated, the cross-sectional nature of this study does not allow the drawing of inferences about causal relationships.

This study also has several strengths. It is one of the few large studies conducted in Saudi Arabia investigating eating disorder pathology and its correlates. The sample size was large, despite several challenges to collect data among Saudi citizens. Saudi Arabia is being perceived as a relatively culturally reclusive society [55]. Therefore, it was difficult to reach Saudi citizens, BM traveled the country to recruit participants. In addition, Saudi Arabia has to deal with stigma in relation to mental health problems, and a lack of knowledge about eating disorders [88–90]. Therefore, study participation in this study was not self-evident for Saudi citizens, as feelings of mistrust, lack of knowledge, and lack of awareness were barriers for study participation in other studies [91-93]. Since female adolescents were overrepresented in the sample results can be potentially be generalized to the eating disorder population as eating disorders are most prevalent among female adolescents [94, 95]. The hypothesis regarding the covariation role of acculturative stress among Saudi citizens with a Western orientation in the development of eating disorder pathology was not confirmed. However, though many studies assume the role of Western orientation and acculturative stress in the development of eating disorder pathology but do not actually measure both constructs, this study measured both. It is also the first study in Saudi Arabia to use a validated eating disorder screening instrument. Most other selfreport measures used in this study also have strong psychometric properties and have been used in other Arab populations before. Furthermore, while most studies assume in- group homogeneity based on nationality, this study examined within-group variance by means of either Western or Arab orientation.

Future research should take into account the limitations of this research, in order to draw causal conclusions, experimental designs should be used. Relating specifically to the proposed implications of this study, causal studies could estimate whether body dissatisfaction and high BMI cause eating disorder pathology or whether they are a result of eating disorder pathology. These studies should use assessor based data regarding BMI to prevent bias [19]. Future research should also investigate if an extensive focus on body dissatisfaction in therapy is actually effective in terms of reduction in eating disorder pathology and relapse rate. In addition, the association between eating disorder pathology and media use should be investigated by emotional involvement rather than time spent or number of actions on (social) media [26, 87] and the use of a validated ARMSA II is recommended. Furthermore, examination of correlates and risk factors of eating disorder pathology is more reliable among clinical samples. However, eating disorders are rarely recognized and treated in Saudi clinics [16, 88]. Besides, Saudi citizens suffering from eating disorders generally seek psychiatric help only after suffering from somatic complaints such as diabetes, infertility, and kidney failure [19]. In addition, although Saudi citizens are an understudied population, a more balanced community sample regarding age, and educational level would potentially increase generalizability.

In conclusion, this explorative study contributes to the current state of knowledge on eating disorder pathology and its correlates in Saudi Arabia. It indicates that BMI is a covariate in the association between eating disorder pathology and body dissatisfaction, and that Saudi citizens with a higher BMI display more severe body dissatisfaction and eating disorder pathology.

#### Abbreviations

ARMSA II: Acculturation Scale for Mexican Americans II; BMI: Body Mass Index; BSQ: Body Shape Questionnaire; EDE-Q: Eating Disorder Examination Questionnaire; PNU: Princess Noura bint Abdulrahman University; PSS: Perceived Stress Scale; SES: Socioeconomic status; WHO: World Health Organization.

#### **Supplementary Information**

The online version contains supplementary material available at https://doi. org/10.1186/s40337-022-00652-4.

Additional file 1. Appendix A. Supplementary Table A. Media use.

#### Acknowledgements

We wish to express our gratitude to all respondents who participated in this study. We gratefully acknowledge the help of Ian Charles Whyte who served as an English corrector.

#### Author contributions

The manuscript has been written by BM in collaboration with MB, EvF and EdB. BM was responsible for cultural adaptation of the measures and data. All authors read and approved the final manuscript. We gratefully acknowledge the help of Sara Abdullah Al Nuwayran and Abeer Ali Al Ghamdi, who participated as native speakers in the translation of the measures and recruited participants. Jeehan Abdulrahman al Qasim also helped recruiting participants.

#### Funding

Not applicable.

#### Availability of data and materials

Upon reasonable request with BM.

#### Declarations

#### Ethics approval and consent to participate

Study approval was given on May 7th, 2017 (17-0097) by the ethical review boards of PNU and King Abulaziz city for Science and Technology, Riyadh, Saudi Arabia. An informed consent was obtained from all participants included in the study.

#### **Competing interests**

All authors declare that there are no competing interests.

#### Author details

<sup>1</sup>Novarum Center for Eating Disorders & Obesity, Laan van de Helende Meesters 2, 1186 AM Amstelveen, The Netherlands. <sup>2</sup>Research Department, Arkin Mental Health Institute, Klaprozenweg 111, 1033 NN Amsterdam, The Netherlands. <sup>3</sup>Section Clinical Psychology, Leiden University, Wassenaarseweg 52, 2333 AK Leiden, The Netherlands. <sup>4</sup>Rivierduinen Eating Disorders Ursula, Sandifortdreef 19, 2333 ZZ Leiden, The Netherlands. <sup>5</sup>Department of Psychiatry, Leiden University Medical Center, Albinusdreef 2, 2333 ZA Leiden, The Netherlands.

Received: 4 March 2022 Accepted: 14 August 2022 Published online: 24 August 2022

#### References

- 1. Gordon RA. Eating disorders East and West: a culture-bound syndrome unbound. Int J Eat Disord. 2001;1:16.
- 2. Pike KM, Hoek HW, Dunne PE. Cultural trends and eating disorders. Curr Opin Psychiatry. 2014;27(6):436–42.
- 3. Pike K, Dunne P. The rise of eating disorders in Asia: a review. J Eat Disord. 2015;3(1):1–14.
- Lewis-Fernández R, Kleinman A. Culture, personality, and psychopathology. J Abnorm Psychol. 1994;103(1):67–71.
- Ford KA, Dolan BM, Evans C. Cultural factors in the eating disorders: a study of body shape preferences of Arab students. J Psychosom Res. 1990;34(5):501–7.
- Thomas J, Khan S, Abdulrahman AA. Eating attitudes and body image concerns among female university students in the United Arab Emirates. Appetite. 2010;54(3):595–8.
- Al Subaie AS. Eating attitude test in Arabic: psychometric features and normative data. Ann Saudi Med. 1999;19:447–53.
- Musaiger AO, Shahbeek NE, Al-Mannai M. The role of social factors and weight status in ideal body shape preferences as perceived by Arab women. J Biosoc Sci. 2004;36(6):699–707.
- 9. Thomas JJ, Lee S, Becker AE. Updates in the epidemiology of eating disorders in Asia and the Pacific. Curr Opin Psychiatry. 2016;29(6):354–62.
- Keel PK, Klump KL. Are eating disorders culture- bound syndromes? Implications for conceptualizing their etiology. Psychol Bull. 2003;129(5):747–69.
- Thomas J, O'Hara L, Quadflieg S, Weissgerber SC. Acculturation, outgroup positivity and eating disorders symptoms among Emirati women. Eat Weight Disord. 2018;23(2):241–6.
- van Son GE, van Hoeken D, Bartelds AI, van Furth EF, Hoek HW. Urbanisation and the incidence of eating disorders. Br J Psychiatry J Mental Sci. 2006;189:562–3.
- Fox JW, Mourtada- SN. The Arab Gulf region: Traditionalism globalized or globalization traditionalized? Globalization and the Gulf. London: Routledge; 2006. p. 13–70.
- Salant T, Lauderdale DS. Measuring culture: a critical review of acculturation and health in Asian immigrant populations. Soc Sci Med. 2003;57(1):71–90.
- Nakai Y, Nin K, Noma S, Teramukai S, Fujikawa K, Wonderlich SA. Changing profile of eating disorders between 1963 and 2004 in a Japanese sample. Int J Eat Disord. 2018;51(8):953–8.
- 16. Melisse B, de Beurs E, van Furth EF. Eating disorders in the Arab world: a literature review. J Eat Disord. 2020;8(1):59.
- Madanat HN, Brown RB, Hawks SR. The impact of body mass index and Western advertising and media on eating style, body image and nutrition transition among Jordanian women. Public Health Nutr. 2007;10(10):1039–46.
- WHO. Country cooperation strategy for WHO and the United Arab Emirates 2005–2009. Geneva: World Health Organisation; 2006.
- Melisse B, van Furth E, de Beurs E. The Eating Disorder Examination-Questionnaire: norms and validity for Saudi Nationals. Eat Weight Disord Stud Anorex Bulim Obes. 2021;27:139–50.
- Qamar Farshori MP, Altamimi N, Alsadery AH, Aljaza ST, Alazmi MS, Muhammad Almahanaa A, et al. Rapidly rising rates of child obesity in schools of hail region of Saudi Arabia: a comparative analysis. Integr Obes Diabetes. 2015;1(6):163–6.
- Villarejo C, Fernández-Aranda F, Jiménez-Murcia S, Peñas-Lledó E, Granero R, Penelo E, et al. Lifetime obesity in patients with eating disorders: increasing prevalence, clinical and personality correlates. Eur Eat Disord Rev. 2012;20(3):250–4.
- 22. Musaiger AO, Al-Mannai M, Tayyem R, Al-Lalla O, Ali EY, Kalam F, et al. Risk of disordered eating attitudes among adolescents in seven Arab countries by gender and obesity: a cross-cultural study. Appetite. 2013;60(1):162–7.
- Maclean EC, Mohammad A, Khatoon N, Abraham S, Carballo M. Perceptions of body size among a multi-ethnic population in Kuwait. Int Diabetes Nurs. 2017;14(2–3):66–71.
- 24. Friedman KE, Reichmann SK, Costanzo PR, Musante GJ. Body image partially mediates the relationship between obesity and psychological distress. Obes Res. 2002;10(1):33–41.

- 25. Derenne J, Beresin EJAP. Body image, media, and eating disorders—a 10-year update. Acad Psychiatry. 2018;42(1):129–34.
- Holland G, Tiggeman M. A systematic review of the impact of the use of social networking sites on body image and disordered eating outcomes. Body Image. 2016;17:100–10.
- 27. Gulf G. Social Media Saudi Arabia statistics and trends. 2017.
- 28. Outlook AM. Arab media exposure. 2012.
- Social Media Penetration in Saudi Arabia Q4 2016. 2017 [cited 19-6-2017]. Available from https://www.statista.com/statistics/284451/saudi-arabiasocial-network-penetration/.
- Stice E, Shaw HE. Role of body dissatisfaction in the onset and maintenance of eating pathology: a synthesis of research findings. J Psychosom Res. 2002;53(5):985–93.
- Roman K. The role of influencers in shaping consumer purchasing decisions in the dietary supplement industry in Poland. Postmod Open. 2020;11(1):93–110.
- 32. Al- Subaie AS. Some correlates of dieting behavior in Saudi schoolgirls. Int J Eat Disord. 2000;28:242–6.
- Thomas J, Quadflieg S, O'Hara L. Implicit out-group preference is associated with eating disorders symptoms amongst Emirati females. Eat Behav. 2016;21:48–53.
- Musaiger AO, Al- MM. Role of obesity and media in body weight concern among female university students in Kuwait. Eat Behav. 2013;14(1):229–32.
- Abdollahi P, Mann T. Eating disorder symptoms and body image concerns in Iran: comparisons between Iranian women in Iran and in America. Int J Eat Disord. 2001;30(3):259–68.
- AlShebali M, AlHadi A, Waller G. The impact of ongoing westernization on eating disorders and body image dissatisfaction in a sample of undergraduate Saudi women. Eat Weight Disord EWD. 2020;26:1835–44.
- Viernes N, Zaidan ZA, Dorvlo AS, Kayano M, Yoishiuchi K, Kumano H, et al. Tendency toward deliberate food restriction, fear of fatness and somatic attribution in cross-cultural samples. Eat Behav. 2007;8(3):407–17.
- Savas E. Do Ramadan fasting restrictions alter eating behaviours in obese women? J Relig Health. 2014;53(1):135.
- 39. Erol A. Do Ramadan fasting restrictions alter eating behaviours? Eur Eat Disord Rev. 2008;16(4):297.
- 40. Ali MM. Holy Quran: Ahmadiyya Anjuman Ishaat Islam Lahore USA; 2011.
- Musaiger A, Al-Awadi A-H, Al-Mannai M. Lifestyle and social factors associated with obesity among the Bahraini adult population. Ecol Food Nutr. 2000;39(2):121–33.
- 42. Lefdahl-Davis EM. The cultural adjustment of Saudi women international students: a qualitative examination. J Cross Cult Psychol. 2015;46(3):406.
- Järvelä-Reijonen E, Karhunen L, Sairanen E, Rantala S, Laitinen J, Puttonen S, et al. High perceived stress is associated with unfavorable eating behavior in overweight and obese Finns of working age. Appetite. 2016;103:249–58.
- Rathner G, Tury F, Szabo P. Eating disorders after the political changes in the formerly communist Eastern-European countries. Orv Hetil. 2001;142(25):1331–3.
- Al Adawi S, Dorvlo AS, Burke DT, Al Bahlani S, Martin RG, Al- IS. Presence and severity of anorexia and bulimia among male and female Omani and non Omani adolescents. J Am Acad Child Adolesc Psychiatry. 2002;4(9):1124–30.
- 46. Burgoine T, Mackenbach JD, Lakerveld J, Forouhi NG, Griffin SJ, Brage S, et al. Interplay of socioeconomic status and supermarket distance is associated with excess obesity risk: a UK cross-sectional study. Int J Environ Res Public Health. 2017;14(11):1290.
- Mitchison D, Hay PJ. The epidemiology of eating disorders: genetic, environmental, and societal factors. Clin Epidemiol. 2014;6:89–97.
- O'Neil A. Saudi Arabia: age structure from 2010 to 2020: Statista; 2022. Available from https://www.statista.com/statistics/262478/age-struc ture-in-saudi-arabia/.
- Worldometers. Saudi Arabia Population Worldometers 2021. Available from https://www.worldometers.info/world-population/saudi-arabiapopulation/.
- Mirkim B. Population levels, trends and policies in the Arab region. Challenges and opportunities. Arab Human Development report. United Nations Development Programme; 2012. 17-09-2016.

- Eapen VM, Mabrouk AA, bin Othman S. Disordered eating attitudes and symptomatology among adolescent girls in the United Arab Emirates. Eat Behav. 2006;7(1):53–60.
- Raouf M, Ebrahimi H, Jafarabadi MA, Malek A, Kheirrodin JB. Prevalence of eating disorders among adolescents in the Northwest of Iran. Iran Red Crescent Med J. 2015;17(10):1–5.
- Gerbasi ME, Richards LK, Thomas JJ, Agnew-Blais JC, Thompson-Brenner H, Gilman SE, et al. Globalization and eating disorder risk: peer influence, perceived social norms, and adolescent disordered eating in Fiji. Int J Eat Disord. 2014;47(7):727–37.
- Brechan I, Kvalem IL. Relationship between body dissatisfaction and disordered eating: mediating role of self-esteem and depression. Eat Behav. 2015;17:49–58.
- Al-Darmaki FR. Attitudes towards seeking professional psychological help: what really counts for United Arab Emirates university students. Soc Behav Personal Int J. 2003;31(5):497–508.
- Fairburn CG, Beglin SJ. Eating Disorder Examination-Questionnaire (6.0). 2008.
- 57. Cooper PJ, Taylor MJ, Cooper Z, Fairbum CG. The development and validation of the Body Shape Questionnaire. Int J Eat Disord. 1987;6(4):485–94.
- Rosen JC, Jones A, Ramirez E, Waxman S. Body shape questionnaire: studies of validity and reliability. Int J Eat Disord. 1996;20(3):315–9.
- Mousa TY, Mashal RH, Al-Domi HA, Jibril MA. Body image dissatisfaction among adolescent schoolgirls in Jordan. Body Image. 2010;7(1):46–50.
- Rosenberg M. Rosenberg self-esteem scale (RSE). Acceptance and commitment therapy. Meas Package. 1965;61(52):18.
- Abdel-Khalek AM. Love of life as a new construct in the well-being domain. Soc Behav Personal Int J. 2007;35(1):125–34.
- 62. Al-Fayez GA, Ohaeri JU, Gado OM. Prevalence of physical, psychological, and sexual abuse among a nationwide sample of Arab high school students: association with family characteristics, anxiety, depression, self-esteem, and quality of life. Soc Psychiatry Psychiatr Epidemiol Int J Res Soc Genet Epidemiol Mental Health Serv. 2012;47(1):53–66.
- 63. Kazarian SS. Arabic contingencies of self worth: Arabic translation and validation of the contingencies of self-worth scale in Lebanese youth. Arab J Psychiatry. 2009;20(2):123–34.
- 64. Cabassa LJ. Measuring acculturation: where we are and where we need to go. Hisp J Behav Sci. 2003;25(2):127–46.
- 65. Cuellar I. Acculturation Rating Scale for Mexican Americans—II: a revision of the original ARSMA scale. Hisp J Behav Sci. 1995;17(3):275–304.
- Stigler M, Dhavan P, Van Dusen D, Arora M, Reddy KS, Perry CL. Westernization and tobacco use among young people in Delhi, India. Soc Sci Med. 2010;71(5):891–7.
- 67. Cohen S, Kamarck T, Mermelstein R. Perceived stress scale: measuring stress. A guide for health and social scientists. Menlo Park: Mind Garden, Inc; 1994. p. 10.
- Almadi T, Cathers I, Hamdan Mansour AM, Chow CM. An Arabic version of the perceived stress scale: translation and validation study. Int J Nurs Stud. 2012;49(1):84–9.
- Fardouly J, Vartanian LR. Negative comparisons about one's appearance mediate the relationship between Facebook usage and body image concerns. Body Image. 2014;12:82–8.
- WHO. Body Mass Index (BMI) related indicators 2021. Available from https://www.who.int/data/gho/data/themes/topics/topic-details/GHO/ body-mass-index.
- 71. Nieva FO. Social women entrepreneurship in the Kingdom of Saudi Arabia. J Glob Entrep Res. 2015;5(1):1.
- 72. General Authority for Statistics SA. Chapter 04 Health. 2016.
- Habibi N. Is Saudi Arabia training too many graduates? 2015. Available from https://www.universityworldnews.com/post.php?story=20150 714013422488.
- 74. Economics T. Saudi Arabia Unemployment Rate Trading Economics 2019.
- Schulte SJ, Thomas J. Relationship between eating pathology, body dissatisfaction and depressive symptoms among male and female adolescents in the United Arab Emirates. Eat Behav. 2013;14(2):157–60.
- ALAhmari T, Alomar AZ, ALBeeybe J, Asiri N, ALAjaji R, ALMasoud R, et al. Associations of self-esteem with body mass index and body image among Saudi college-age females. Eat Weight Disord EWD. 2019;24(6):1199–207.
- Unikel Santoncini C, Bojorquez Chapela I, Díaz de León Vázquez C, Vázquez Velázquez V, Rivera Márquez JA, Galván Sánchez G, et al.

Validation of eating disorders examination questionnaire in Mexican women. Int J Eating Disord. 2018;51(2):146–54.

- Ghubash R, Daradkeh TK, Al-Muzafari SMA, El-Manssori ME, Abou-Saleh MT. Al-Ain community psychiatric survey IV: socio-cultural changes (traditionality-liberalism) and prevalence of psychiatric disorders. Soc Psychiatry Psychiatr Epidemiol. 2001;36(11):565–70.
- Sato K. Who is happier in Japan, a housewife or working wife? J Happ Stud. 2022;23(2):509–33.
- Mond JM, Hay PJ, Rodgers B, Owen C. Eating Disorder Examination Questionnaire (EDE-Q): norms for young adult women. Behav Res Ther. 2006;44(1):53–62.
- Rø Ø, Reas DL, Rosenvinge J. The impact of age and BMI on Eating Disorder Examination Questionnaire (EDE-Q) scores in a community sample. Eat Behav. 2012;13(2):158–61.
- Gorber SC, Tremblay M, Moher D, Gorber B. A comparison of direct versus self-report measures for assessing height, weight and body mass index: a systematic review. Obesity Rev. 2007;8(4):307–26.
- Stice E, Shaw H, Marti CN. A meta-analytic review of eating disorder prevention programs: encouraging findings. Annu Rev Clin Psychol. 2007;3:207–31.
- Bar RJ, Cassin SE, Dionne MM. The long-term impact of an eating disorder prevention program for professional ballet school students: a 15-year follow-up study. Eat Disord. 2017;25(5):375–87.
- Kindermann S, Moessner M, Ozer F, Bauer S. Associations between eating disorder related symptoms and participants' utilization of an individualized Internet-based prevention and early intervention program. Int J Eat Disord. 2017;50(10):1215–21.
- Stice E, Johnson S, Turgon R. Eating disorder prevention. Psychiatr Clin North Am. 2019;42(2):309–18.
- Fardouly J, Vartanian LR, Diedrichs PC, Halliwell E. Social comparisons on social media: the impact of Facebook on young women's body image concerns and mood. Body Image. 2015;13(C):38–45.
- Alkhadari S, Alsabbrri AO, Mohammad IHA, Atwan AA, Alqudaihi F, Zahid MA. Prevalence of psychiatric morbidity in the primary health clinic attendees in Kuwait. J Affect Disord. 2016;195:15–20.
- Latzer Y, Azaiza F, Tzischinsky O. Eating attitudes and dieting behavior among religious subgroups of Israeli–Arab adolescent females. J Relig Health. 2009;48(2):189–99.
- Griffiths S, Mond JM, Murray SB, Thornton C, Touyz S. Stigma resistance in eating disorders. Soc Psychiatry Psychiatr Epidemiol. 2015;50(2):279–87.
- Williams MT, Beckmann-Mendez DA, Turkheimer E. Cultural barriers to African American participation in anxiety disorders research. J Natl Med Assoc. 2013;105(1):33–41.
- 92. Schmotzer GL. Barriers and facilitators to participation of minorities in clinical trials. Ethn Dis. 2012;22(2):226–30.
- Honein-AbouHaidar GN, Kastner M, Vuong V, Perrier L, Daly C, Rabeneck L, et al. Systematic review and meta-study synthesis of qualitative studies evaluating facilitators and barriers to participation in colorectal cancer screening. Cancer Epidemiol Biomark Prev Publ Am Assoc Cancer Res Cosponsored Am Soc Prev Oncol. 2016;25(6):907–17.
- 94. Phillipou A, Meyer D, Neill E, Tan EJ, Toh WL, Van Rheenen TE, et al. Eating and exercise behaviors in eating disorders and the general population during the COVID-19 pandemic in Australia: initial results from the COL-LATE project. Int J Eat Disord. 2020;53(7):1158–65.
- Mitchison D, Mond J, Bussey K, Griffiths S, Trompeter N, Lonergan A, et al. DSM-5 full syndrome, other specified, and unspecified eating disorders in Australian adolescents: prevalence and clinical significance. Psychol Med. 2020;50(6):981–90.

#### **Publisher's Note**

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

#### Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

#### At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

